Author's response to reviews

Title: The Morbidity of Urethral Stricture Disease among Male Medicare Beneficiaries

Authors:

Jennifer T Anger (janger@mednet.ucla.edu)
Richard Santucci (RSantucc@dmc.org)
Anna L Grossberg (annleah@ucla.edu)
Christopher S Saigal (CSaigal@mednet.ucla.edu)

Version: 2 Date: 12 November 2009

Author's response to reviews: see over
Dear Dr. Norton,

Thank you for reviewing our original manuscript entitled “THE MORBIDITY OF URETHRAL STRICTURE DISEASE AMONG MALE MEDICARE BENEFICIARIES.” We most appreciate the reviewers’ comments. We have revised the manuscript based on the editor’s recommendations, as outlined below.

1. We describe that this study obtained an IRB exemption in the methods section.

2. We revised the background section of the abstract, as recommended.

3. We include a competing interests section.

4. We added a author contributions section.

We very much appreciate your time and consideration.

Sincerely,

Jennifer Anger, MD
Corresponding Author
Assistant Professor of Urology
UCLA-Santa Monica
1260 15th Street, Suite 1200
Santa Monica, CA  90404
Tel: (310) 451-8751
FAX: (310) 206-5343
E-mail: janger@mednet.ucla.edu